Author’s response to reviews

Title: Serotonin transporter gene polymorphism may be associated with functional dyspepsia in a Japanese population

Authors:

Fumihiko Toyoshima (f-toyo@hyo-med.ac.jp)
Tadayuki Oshima (t-oshima@hyo-med.ac.jp)
Shigemi Nakajima (shigemi@ares.eonet.ne.jp)
Jun Sakurai (jsakurai@hyo-med.ac.jp)
Junji Tanaka (junji-t@hyo-med.ac.jp)
Toshihiko Tomita (tomita@hyo-med.ac.jp)
Kazutoshi Hori (k-hori@hyo-med.ac.jp)
Takayuki Matsumoto (matsut@hyo-med.ac.jp)
Hiroto Miwa (miwahgi@hyo-med.ac.jp)

Version: 5 Date: 17 June 2011

Author’s response to reviews: see over
Dear the Editor,

Thank you for accepting our manuscript (MS: 1469429308471213) “Serotonin transporter gene polymorphism may be associated with functional dyspepsia in a Japanese population”.

Here is a point-by-point description of the changes made.

# We added street name of institutes and corresponding author information.
# We erased red-colored revised points which were used for the reviewers.
# We included the tables in the manuscript (PDF).

Thank you for handling our manuscript.

Sincerely,

Hiroto Miwa, M.D., Ph.D.
Division of Upper Gastroenterology
Department of Internal Medicine
Hyogo College of Medicine
1-1 Mukogawa-cho, Nishinomiya, Hyogo 663-8501, Japan
TEL: +81-798-45-6665
FAX: +81-798-45-6661
E-mail: miwahgi@hyo-med.ac.jp